# The Change in Living Kidney Donation in Women and Men in the United States (2005–2015): A Population-Based Analysis

Jagbir Gill,<sup>1,2</sup> Yayuk Joffres,<sup>1</sup> Caren Rose,<sup>1</sup> Julie Lesage,<sup>3</sup> David Landsberg,<sup>1</sup> Matthew Kadatz,<sup>1</sup> and John Gill<sup>1,2,4</sup>

<sup>1</sup>Division of Nephrology, University of British Columbia, Vancouver, British Columbia, Canada; <sup>2</sup>Department of Medicine, Center for Health Evaluation and Outcomes Sciences, Vancouver, British Columbia, Canada; <sup>3</sup>Department of Medicine, Centre Hospitalier de l'Universite de Montreal, Montréal, Québec, Canada; and <sup>4</sup>Department of Medicine, Tufts–New England Medical Center, Boston, Massachusetts

### ABSTRACT

The factors underlying the decline in living kidney donation in the United States since 2005 must be understood to inform strategies to ensure access to this option for future patients. Population-based estimates provide a better assessment of donation activity than do trends in the number of living donor transplants. Using data from the Scientific Registry of Transplant Recipients and the United States Census, we determined longitudinal changes in living kidney donation between 2005 and 2015, focusing on the effect of sex and income. We used multilevel Poisson models to adjust for differences in age, race, the incidence of ESRD, and geographic factors (including population density, urbanization, and daily commuting). During the study period, the unadjusted rate of donation was 30.1 and 19.3 per million population in women and men, respectively, and the adjusted incidence of donation was 44% higher in women (incidence rate ratio [IRR], 1.44; 95% confidence interval [95% CI], 1.39 to 1.49). The incidence of donation was stable in women (IRR, 0.95; 95% CI, 0.84 to 1.07) but declined in men (IRR, 0.75; 95% CI, 0.68 to 0.83). Income was associated with longitudinal changes in donation in both sexes, yet donation was stable in the highest two population income quartiles in women but only in the highest income quartile in men. In both sexes, living related donations declined, irrespective of income. In conclusion, living donation declined in men but remained stable in women between 2005 and 2015, and income appeared to have a greater effect on living donation in men.

J Am Soc Nephrol 29: 1301–1308, 2018. doi: https://doi.org/10.1681/ASN.2017111160

Living donor kidney transplantation is the preferred treatment for patients with ESRD.<sup>1</sup> The number of living donor transplantations in the United States has been decreasing since 2005,<sup>2</sup> for uncertain reasons,

Received November 6, 2017. Accepted February 17, 2018.

Published online ahead of print. Publication date available at www.jasn.org.

**Correspondence:** Dr. Jagbir Gill, Division of Nephrology, University of British Columbia, St. Paul's Hospital, Providence Building Ward 6a, 1081 Burrard Street, Vancouver, BC V6Z 1YK, Canada. Email: jsgill2@me.com

Copyright © 2018 by the American Society of Nephrology

### Significance Statement

Living donor kidney transplantation is the preferred treatment for patients with ESRD, but the number of living donor transplantations in the United States has been decreasing since 2005, for uncertain reasons. Using data from the Scientific Registry of Transplant Recipients and the United States Census, we determined longitudinal changes in living donation between 2005 and 2015, focusing on the effect of sex and income. Living donation declined in men but remained stable in women; income appeared to have a greater effect on living donation in men. These findings suggest that strategies to remove financial barriers to living donation may reduce gender based disparities in donation. and studies are needed to better understand the factors contributing to this decline so that evidence-based strategies may be implemented to ensure patients with ESRD continue to benefit from this treatment in the future.<sup>2</sup>

Women comprise the majority of living kidney donors in the United States, and accounted for 63% of all living donors in 2016.<sup>3</sup> The reasons underlying the imbalance in living donation between women and men are also incompletely understood. It has been hypothesized that a higher incidence of kidney disease, heart disease, diabetes, and hypertension in men may limit their suitability for donation.<sup>4–6</sup> In addition, the financial implications of donating may pose a larger barrier for living donation in men compared with women because more men are likely to be the primary household earner.<sup>7</sup>

Simply reporting the number of living donor transplantations in national registries can provide an incomplete understanding of living donor activity because these analyses do not account for population-level differences in factors such as the prevalence of ESRD and income, which may affect the number of living donor transplantations. In contrast, population-based analyses that account for differences in factors that can affect the number of living donor transplantations provide a more direct assessment of living donor activity, allow for determination of longitudinal changes in living donation in a given country or region, and facilitate comparisons between countries or regions. For example, the lower number of living donor transplantations in black people<sup>8,9</sup> has led to presumption that they have a low level of support for living donation.<sup>10,11</sup> However, in a population-based analysis that accounted for race-related differences in ESRD and income, black people were shown to have a higher rate of living donation per million population compared with white people, even after accounting for differences in ESRD rates.12

To better understand the contemporary decline in living donation in the United States over the past decade, we undertook a population-level analysis of living kidney donation, focusing on longitudinal changes since 2005 and the effect of donor sex and income. Given our previous work showing the association of income with living donation,<sup>13</sup> we hypothesized a greater decline in living donation in men compared with women, and that sex-related differences in living donation would be greater in lower income groups.

#### RESULTS

Figure 1 outlines the cohort selection of living donors. After excluding n=490 living kidney donors who were not between the ages of 18 and 69 years and n=8531 donors for whom income could not be determined because of missing residential zip code data, n=52,690 living donors (85.4% of all living donors in the United States during the study period of 2005–2015) including n=31,958 women and n=20,732 men) were identified for study inclusion. The characteristics of the 8531 donors excluded because of missing zip code data were similar to those of the study cohort (data not shown).



Figure 1. Cohort selection.

Table 1 outlines the characteristics of women and men who were living kidney donors in the United States between 2005 and 2015. Both women and men donated mostly to male recipients, with 63% of women and 59% of men donating to men. Compared with men, more women donated to spouses and fewer were living related donors. There were proportionately fewer women donors in the age group 18–34 years, whereas there were small but statistically significant differences in race between women and men (11% of women

Table 1.Characteristics of living kidney donors by sex in theUnited States between 2005 and 2015

Chamadaniatian	Women	Men	
Characteristics	n=31,958	n=20,732	Pvalue
Age categories, %			< 0.001
18–34	27	35	
35–49	44	41	
50–69	29	24	
Race, %			< 0.001
Black	11	12	
White	71	70	
Other	18	18	
Donor relationship, <sup>b</sup> %			
Living related	53	60	
Living unrelated	31	32	< 0.001
Spousal	16	9	
Median household income			
Quartiles ( <i>N</i> ), %			
Q1 (<\$39,287)	15 (4793)	16 (3317)	0.21
Q2 (\$39,287–\$49,120)	20 (6392)	20 (4146)	
Q3 (\$49,121–\$61,902)	24 (7670)	24 (4976)	
Q4 (>\$61,902)	41 (13,103)	40 (8393)	
RUCA, %			< 0.001
>50,000	82	83	
10,000–50,000	9	9	
<10,000	9	8	
Recipient sex, %			
Women	37	41	< 0.001
Men	63	59	

<sup>a</sup>Where not stated, missing values comprised <2% of all values. <sup>b</sup>Donor income categories on the basis of data from the 2010 United States Census.



**Figure 2.** Unadjusted living donor rates per million population (pmp) in women and men, stratified by donor relation between 2005 and 2015 and adjusted IRRs of living donation in women compared with men.

were black compared with 12% of men) and minor differences in rural–urban commuting area (RUCA) distribution (82% of women versus 83% of men lived in metropolitan areas).

Supplemental Table 1 outlines the characteristics of women and men aged 18–69 years, as reported in the 2010 United States Census. Compared with men, women were slightly older and included a larger proportion of black people. Median household income was similar between women and men in the United States population and had minor statistically different RUCA distributions (81% of women versus 80% of men lived in metropolitan areas). The period prevalent rate of ESRD (calculated using US Renal Data System [USRDS] and 2010 United States Census data between 2000 and 2015) was significantly higher in men, at a rate of 394 men per million population compared with 253 per million population in women (P<0.001).

# Rates of Living Donation in Women and Men during the Period 2005–2015

Figure 2 shows the unadjusted rates of living donation in women and men during the study period and the adjusted relative incidence of donation in women versus men (labeled incidence rate ratio [IRR] in the figure), stratified by the donor recipient relationship. The unadjusted rate of living donation in women was 1.5 times that of men, with a rate of 30.1 per million population in women compared with 19.3 per million population in men. After adjustment for age, race, median household income, rate of ESRD, and RUCA, women had a 44% higher incidence of donation compared with men (IRR, 1.44; 95% confidence interval [95% CI], 1.39 to 1.49). This difference was greatest for spousal donation, where women had a 148% higher adjusted incidence of donation compared with men (IRR, 2.48; 95% CI, 2.33 to 2.65), whereas women had a 31% higher adjusted incidence of living related donation (IRR, 1.31; 95% CI, 1.25 to 1.36) and a 44% higher adjusted incidence of unrelated donation (IRR, 1.44; 95% CI, 1.36 to 1.52).

# Longitudinal Changes in Living Kidney Donation in Men and Women

Figure 3 shows the unadjusted rate of living donation per million population per year and the adjusted relative incidence of donation (IRR) during the period 2005–2015 in women and men, separately. The rate of living donation in women was stable during the 10-year study period, with no significant change in the adjusted incidence of living donation (IRR, 0.95; 95% CI, 0.84 to 1.07). In contrast, the unadjusted donation rate in men declined from 21.9 per million population in 2005 to 16.8 per million population in 2015, with a 25% decline in the incidence of donation after adjustment for age, race, ESRD rate, income, and RUCA (IRR, 0.75; 95% CI, 0.68 to 0.83). The change in living donation over time was statistically different in women compared with men (P value for interaction <0.001).

Figure 4 displays the unadjusted rate of living donation in women (A) and men (B), stratified by zip code level population quartiles of median household income. Among women (Figure 4A), the unadjusted rate of donation was greater in higher income populations and the decline in donation over time varied by median household income. Between 2005 and 2015, the unadjusted rate of living donation was stable in the two highest income quartiles (i.e., quartiles 3 and 4) and declined in the two lowest income quartiles. Table 2 shows the adjusted relative incidence of living kidney donation between 2005 and 2015, stratified by zip code level median household income quartiles. Living donation declined by 27% among women in the lowest income quartile (Q1) and by 16% in the second lowest income quartile (Q2), but the adjusted relative incidence of donation was stable in the top two quartiles (Q3 and Q4) of median household income, resulting in an overall stable donation rate during the study period among women (IRR, 0.95; 95% CI, 0.84 to 1.07).

In contrast, men had a more marked decline in both the unadjusted and adjusted rate of living donation between 2005 and 2015 (Figure 4B, Table 2). The unadjusted rate of living donation remained stable only among men in the highest



**Figure 3.** Living donor rates per million population per year (pmp/yr) between 2005 and 2015 in women and men.



**Figure 4.** Living donor rates per million population per year (pmp/yr) between 2005 and 2015 in (A) women and (B) men, stratified by quartile of median household income. LD, living donor.

income quartile (Q4), and declined among men in the three lower income quartiles (Figure 4B). The adjusted relative incidence of living kidney donation in men declined by 25% overall between 2005 and 2015 (IRR, 0.75; 95% CI, 0.68 to 0.83), and declined by 41% in Q1, 40% in Q2, and 29% in Q3 between 2005 and 2015 (Table 2). There was no change in the adjusted relative incidence of living kidney donation during the study period among men in the highest income quartile (Q4). There was a significant interaction (P=0.05) between income and sex for the outcome of living donation during the study period.

Table 3 shows the adjusted relative incidence of donation during the study period in women and men, stratified by the donor and recipient relationship. Among women, only living related donations declined during the study period 2005–2015 (IRR, 0.68; 95% CI, 0.60 to 0.76). In contrast, among men the adjusted relative incidence of donation during the study period 2005–2015 declined in every donor recipient relationship subgroup: the largest decline (47%) was observed in living related donations, followed by spousal donations (22%), and unrelated donation (17%).

Given the large longitudinal decline in living related donation among both women and men, an analysis of the change in

Table 2.Longitudinal change in the adjusted relativeincidence of living kidney donation 2005 and 2015 in womenand men, stratified by median household income

Madian Hawada dalah sama	Women	Men
iviedian Housenoid Income	IRR (95% CI)	IRR (95% CI)
All	0.95 (0.84 to 1.07)	0.75 (0.68 to 0.83) <sup>a</sup>
Q1 (<\$39,287)	0.73 (0.60 to 0.87) <sup>a</sup>	0.59 (0.52 to 0.68) <sup>a</sup>
Q2 (\$39,287–\$49,120)	0.84 (0.72 to 0.97) <sup>a</sup>	0.60 (0.53 to 0.68) <sup>a</sup>
Q3 (\$49,121–\$61,902)	0.95 (0.84 to 1.09)	0.71 (0.60 to 0.84) <sup>a</sup>
Q4 (>\$61,902)	1.11 (0.96 to 1.28)	0.96 (0.86 to 1.07)

IRR refers to the relative incidence of living kidney donation in 2015 relative to 2005, on the basis of a multilevel Poisson regression model adjusted for donor- and population-level age, race, ESRD rate standardized for race and age, and RUCA, clustered by state of residence. For example, men in the lowest income quintile (Q1) had a 41% decline in living donation between 2005 and 2015, with an IRR of 0.59.

living related donation by income quartile was undertaken (Table 4). The adjusted relative incidence of living related donations dropped in all income quartiles in both men and women.

Analyses of longitudinal changes in spousal and unrelated donations by income quartile in men and women showed the same pattern as observed in the overall analysis: among women, the longitudinal decline in spousal and unrelated donations was confined to the lowest two income quartiles, whereas among men, spousal and unrelated donations declined in the three lowest income quartiles during the study period (data not shown).

#### DISCUSSION

This study represents the first population-level analysis of living kidney donation in women and men in the United States in the current era, when overall living donation rates have declined and during an economically strained time in the United States, with increased rates of unemployment and decreased job security.<sup>14</sup> We found that women have a 44% higher incidence of living kidney donation compared with men, after

**Table 3.** Longitudinal change in the adjusted relative incidence of living kidney donation during the period 2005 and 2015 in women and men, stratified by donor relation

Dowow Dolotion	Women	Men
Donor Relation	IRR (95% CI)	IRR (95% CI)
<b>Д</b>	0.95 (0.84 to 1.07)	0.75 (0.68 to 0.83) <sup>a</sup>
Spousal	0.91 (0.75 to 1.09)	0.78 (0.62 to 0.99) <sup>a</sup>
Related	0.68 (0.60 to 0.76) <sup>a</sup>	0.53 (0.48 to 0.59) <sup>a</sup>
Unrelated	1.02 (0.88 to 1.18)	0.83 (0.71 to 0.96) <sup>a</sup>

IRR refers to the relative incidence of living kidney donation in 2015 relative to 2005, on the basis of a multilevel Poisson regression model adjusted for donor- and population-level age, race, ESRD rate standardized for race and age, and RUCA, clustered by state of residence. Overall, men had a 25% decline in living donation between 2005 and 2015, with an IRR of 0.75. <sup>a</sup>P<0.05.

Table 4.	Longitudinal change in the adjusted relative
incidence	of living related kidney donation during the period
2005 and	2015 in women and men

Zip Code Level	Women	Men
Income Quartiles	IRR (95% CI)	IRR (95% CI)
Q1 (<\$39,287)	0.55 (0.45 to 0.68) <sup>a</sup>	0.44 (0.37 to 0.51) <sup>a</sup>
Q2 (\$39,287–\$49,120)	0.65 (0.56 to 0.76) <sup>a</sup>	0.43 (0.36 to 0.51) <sup>a</sup>
Q3 (\$49,121–\$61,902)	0.64 (0.57 to 0.73) <sup>a</sup>	0.49 (0.41 to 0.60) <sup>a</sup>
Q4 (>\$61,902)	0.78 (0.67 to 0.90) <sup>a</sup>	0.69 (0.61 to 0.78) <sup>a</sup>

IRR refers to the relative incidence of living kidney donation in 2015 relative to 2005, on the basis of a multilevel Poisson regression model adjusted for donor- and population-level age, race, ESRD rate standardized for race and age, and RUCA, clustered by state of residence. For example, women in the lowest income quintile (Q1) had a 45% decline in living related donation between 2005 and 2015, with an IRR of 0.55.

<sup>a</sup>P<0.05.

adjustment for factors that have been previously hypothesized to explain this disparity, including sex-specific differences in ESRD. We also found that living donation in women has remained stable over the last decade, whereas there has been a marked decrease in living donation in men, resulting in a widening gap in donation between the sexes. We found that living donation rates varied by income in both women and men, but the decline in living donation was most marked in men from lower income groups. Finally, we found that living related donations declined in both women and men, in all income groups.

The last national population-based analysis examining sex differences in living donation was on the basis of transplantations performed during the years 1991–1993.<sup>15</sup> Our analysis is timely because it examines sex differences in donation during a period when living donation is declining. Although it is difficult to compare the results from the 1990s to our findings, the fact that we demonstrate a widening of the discrepancy in donation between women and men over the study period should renew interest in understanding the basis for this long-standing discrepancy. Importantly, the previous work did not include consideration of sex related differences in income or ESRD.<sup>15</sup> Our finding that sex-related differences in living donation persisted after adjustment for differences in ESRD between women and men suggests the importance of other factors including differences in the financial implications of donation, attitudes toward donation, or societal expectations of donation between women and men contribute to the sex-related difference in living donation.<sup>5,15</sup>

Our findings also draw further attention to financial barriers to living donation (see recommendations from the American Society of Transplantation Living Donor Community of Practice consensus conference<sup>16</sup> and other recent publications<sup>17,18</sup> for a comprehensive overview of this issue). The finding that income had a greater effect on donation from men should be interpreted with the knowledge that living donors and recipients are usually from the same socioeconomic background and, in many cases, the same household.<sup>19</sup> It is notable that the largest difference (IRR, 2.48) in donation between women and men was observed for spousal donations. Because a larger proportion of men are the primary household income earner in the United States<sup>20</sup> and are likely to earn higher wages compared with women,<sup>7</sup> it follows that the potential for lost wages may disproportionately affect donations from men. With the exception of selected private charities and paired exchange registries, most current expense reimbursement programs for living donors do not cover lost wages. The National Living Donor Assistance Center does not provide support for lost wages, but it is currently engaged in a randomized trial in which potential transplant recipients will be randomized to be offered or not offered wage reimbursement for their donor.<sup>21</sup>

It is important to note that financial barriers to donation are not limited to out of pocket expenses and lost wages. Concerns about financial, employment, or insurance consequences of donation may have been particularly relevant in our analysis, as it was conducted during the period of economic downturn in the United States. The potential for an interruption in employment or loss of employment may pose a greater concern for loss of work-related benefits, including health insurance, in men. Ensuring adequate medical follow-up for living donors postdonation is increasingly recognized a priority for transplantation programs,<sup>22,23</sup> and therefore lack of health insurance may be a significant barrier to donation. Because a larger proportion of women in the United States have their health insurance benefits covered as a dependent,24 the threat of job loss may not necessarily affect their ability to access health care in the future. Conversely, the prospect of a male donor losing his job may more frequently result in loss of health insurance for him and his dependents. The proposed Living Donor Protection Act<sup>25</sup> would entitle a covered employee to medical leave in the event of living kidney donation. The bill would also prohibit discrimination on the basis of an individual's status as a living organ donor in "the offering, issuance, cancellation, coverage, price, or any other condition of a life insurance policy, disability insurance policy, or long-term care insurance policy." Our findings suggest that the passage of this bill into law may indirectly help increase living donation in the future, particularly from men.

Although our study emphasizes the importance of income as a barrier to donation in men, it is important to highlight that income was associated with lower donation rates in both women and men. We previously reported that the retraction in living donation since 2005 was restricted to lower income populations.<sup>13</sup> Together, these findings highlight the need for broad based policies to ensure donation is financially neutral for living donors.<sup>26,27</sup> This includes reimbursement for out of pocket expenses at the time of donation, and for long-term health insurance coverage for donation related complications, as well as for health care required to preserve postdonation kidney function.<sup>28</sup>

Our findings also highlight the role of additional factors beyond financial considerations in the decline in living donation over the past decade. We found that living related donation declined by 42% in women and 47% in men between 2005 and 2015. Of note, this decline did not vary by median household

#### CLINICAL EPIDEMIOLOGY www.jasn.org

income. Parent to child living donations have declined, in part, because of changes in deceased donor organ allocation that prioritize pediatric recipients,29 but this only constitutes a minority of living related donations. Studies about the long-term health outcomes of living donors were published in 2013<sup>30</sup> and 2014,<sup>31</sup> and may have had some limited effect on the use of living related donations in the latter years of our analysis, but were unlikely to contribute to the decline in donation that preceded these studies. It is important to understand that other than the potential role of APOL-1 gene variants,<sup>32–36</sup> the basis for an increased risk of kidney failure in living related donors remains uncertain. Importantly, lifestyle factors that are shared among family members may lead to conditions such as diabetes and hypertension, which appear to be important antecedents of CKD in previous living donors.<sup>37,38</sup> The importance of these recent observations is that much of the apparent familial risk in related living donors may be modifiable with appropriate postdonation care.

The strengths of this analysis are that it includes all living donors in the United States with a valid zip code and directly examined prevalent and longitudinal changes in living donor transplantation in women and men, after adjustment for differences in age, race, geography, and ESRD. Our populationlevel analysis may not be applicable to individual donors. The use of zip codes to determine median household income is frequent in the medical literature and assumes the same income for individuals living in a given zip code. This assumption may be incorrect, especially in metropolitan areas. Median household income is only one indicator of socioeconomic status and may not directly relate to the financial status of an individual. However, when we examined other metrics at the zip code level, such as education, we found that these were colinear with median household income. Because we examined median household income as a categorical variable rather than a continuous variable, we were unable to account for the variability in income within each income quintile and how this relates to living donation. Data from the 2010 United States Census were used to assign median household income and determine population figures. It is important to note that the lack of individual level data limit our ability to account for changes in median household income over time, particularly during periods of economic stress. We adjusted for the age and race standardized prevalence of ESRD, but are not able to directly adjust for the incidence of other health conditions (i.e., diabetes, heart disease) that may affect living donation. The effect of such conditions would only be accounted for to the extent that they are correlated with the prevalence of ESRD in our study. We have previously reported on the equivalent rate of living donation in black and white Americans in the United States, and therefore we did not specifically examine three-way interactions of ancestry, income, and sex on living donation in this study as these analyses are difficult to interpret and we have shown that black ancestry is not an independent determinant of living donation after accounting for differences in income and ESRD. Specific to black people, recent survey work has shown

that United States transplant centers do not routinely currently use APOL-1 testing in their evaluation of living donors,<sup>32</sup> and therefore we do not believe the evolving information regarding APOL-1 risk alleles and the risk of CKD affected the use of living related donation in black people in this study. The trends in this analysis may not be applicable for small selected subgroups of donors, such as nondirected anonymous donors. Kumar *et al.*<sup>39</sup> recently reported on nondirected donation in the United States and showed that only 44% of these donors were men. Although we did not separately examine trends in living donation by sex and income among nondirected anonymous donors in this analysis because of the small numbers of such donors, it is unlikely that these donors were key drivers of the overall trends in donation in women and men, given their small numbers.

In summary, we found that changes in living donation in the past decade varied by sex, resulting in a widened gap in donation between women and men and that income may be an important factor contributing to this discrepancy. These findings suggest that strategies to remove financial barriers to living donation as may be important to maintain living donation in the future. The reasons for the decline in living related donation require further study.

### **CONCISE METHODS**

### Definitions

Living related donations included all donations from a biologic relative including donations between sibling, parent to child donations, and donations from adult offspring to a parent. Living unrelated donations included all nonspousal, nonbiologically related donors.

#### **Data Source and Study Population**

Data from the Scientific Registry of Transplant Recipients (SRTR) were used to identify all living kidney donors aged 18–69 years of age in the United States between 2005 and 2015. The SRTR data system includes data on all donor, waitlist candidates, and transplant recipients in the United States, submitted by the members of the Organ Procurement and Transplant Network (OPTN). The Health Resources and Services Administration, US Department of Health and Human Services provides oversight to the activities of the OPTN and SRTR contractors.

Individuals aged 70 and older were excluded because they represent a very small proportion of living donors and estimates of donation may not be robust in these groups. Data from the 2010 United States Census were used to determine zip code level median household income. Zip codes were then grouped into quartiles on the basis of median household income.

The characteristics of women and men in the United States population in the 2010 United States Census and in living donors between 2005 and 2015 were examined. Continuous variables were reported as mean $\pm$ SD, or medians (25th, 75th percentile), whereas categorical variables were described using proportions. Group differences were determined using *t* tests, ANOVA, or the chi-squared test as appropriate.

# Adjusted Relative Incidence of Living Donation in Women Compared with Men

The overall rate of living kidney donation per million population was determined in women and men during the study period and was stratified by donor relation (spousal, related, and unrelated). Multivariable Poisson regression models were used to determine the adjusted relative incidence of living donation in women versus men during the study period, and was further stratified by donor relation. The models were adjusted for donor- and population-level differences in age, race, and median household income. To account for the possibility that differences in ESRD in women and men may affect living donation, we also adjusted for age and race standardized rate of ESRD in women and men. Zip code level period prevalent ESRD rates were determined using data from USRDS and the 2010 United States Census, using methods we have used in prior analyses.<sup>13</sup> Finally, to account for geographical factors that may affect the likelihood of living donation, all analyses were clustered by state of residence and adjusted for population density within each zip code using RUCA code.40,41 RUCA codes were classified into the following groups: metropolitan (cities with population of >50,000 and their associated suburban areas; RUCA, 1.0-3.9), micropolitan (towns or cities with population 10,000-50,000; RUCA, 4.0-6.0), and rural (towns with a population <10,000; RUCA>6.0).

# Longitudinal Changes in Living Donation in Women and Men

To examine sex-specific changes in donation over time, living kidney donation rates per year were determined separately for women and men between 2005 and 2015. The sex-specific adjusted relative incidence of living donation per year between 2005 and 2015 was determined using multivariable Poisson regression models. These models were adjusted for age, race, median household income, and race and age standardized ESRD rates in women and men. All analyses were clustered by state of residence and adjusted for population density within each zip code using RUCA codes.<sup>40,41</sup> In addition, the interaction between sex and time for the outcome of donation was formally tested.

Data from the 2010 United States Census were used to define zip code quartiles on the basis of median household income. Adjusted sexspecific changes in living donation between 2005 and 2015 were then determined within each income quartile, using the same methods outlined above. These analyses were repeated for subgroups of donation: living related, spousal, and living unrelated donation. To further examine the effect of income on the difference in donation rates between women and men during the study period, the interaction of sex and income for the outcome of donation was formally tested.

This study was conducted with the approval of our local hospital research ethics board. All analyses were performed using SAS v9.4 (SAS Institute, Cary, NC).

### ACKNOWLEDGMENTS

The data reported here have been supplied by the Scientific Registry of Transplant Recipients (SRTR) and the US Renal Data System.

Jagbir Gill is a Scholar of the Michael Smith Foundation for Health Research. John Gill is supported by a Foundation Award from the Canadian Institutes of Health Research.

The interpretation and reporting of these data are the responsibility of the author(s) and in no way should be seen as an official policy of or interpretation by the Organ Procurement and Transplant Network/ SRTR or the US Government.

#### DISCLOSURES

None.

## REFERENCES

- Tonelli M, Wiebe N, Knoll G, Bello A, Browne S, Jadhav D, et al.: Systematic review: Kidney transplantation compared with dialysis in clinically relevant outcomes. Am J Transplant 11: 2093–2109, 2011
- Rodrigue JR, Schold JD, Mandelbrot DA: The decline in living kidney donation in the United States: Random variation or cause for concern? *Transplantation* 96: 767–773, 2013
- 3. Available at: http://www.unos.org. Accessed October 22, 2017
- Kayler LK, Meier-Kriesche HU, Punch JD, Campbell DA Jr., Leichtman AB, Magee JC, et al.: Gender imbalance in living donor renal transplantation. *Transplantation* 73: 248–252, 2002
- 5. Kayler LK, Rasmussen CS, Dykstra DM, Ojo AO, Port FK, Wolfe RA, et al.: Gender imbalance and outcomes in living donor renal transplantation in the United States. *Am J Transplant* 3: 452–458, 2003
- Zimmerman D, Donnelly S, Miller J, Stewart D, Albert SE: Gender disparity in living renal transplant donation. Am J Kidney Dis 36: 534–540, 2000
- U.S. Bureau of Labor Statistics: Available at: https://www.bls.gov. Accessed October 22, 2017
- Weng FL, Reese PP, Mulgaonkar S, Patel AM: Barriers to living donor kidney transplantation among black or older transplant candidates. *Clin J Am Soc Nephrol* 5: 2338–2347, 2010
- Hall EC, James NT, Garonzik Wang JM, Berger JC, Montgomery RA, Dagher NN, et al.: Center-level factors and racial disparities in living donor kidney transplantation. *Am J Kidney Dis* 59: 849–857, 2012
- Ayanian JZ, Cleary PD, Keogh JH, Noonan SJ, David-Kasdan JA, Epstein AM: Physicians' beliefs about racial differences in referral for renal transplantation. Am J Kidney Dis 43: 350–357, 2004
- African Americans less likely to receive a living kidney donation, May 2012. Available at: https://www.kidney.org/news/newsroom/nr/racial\_transplant. Accessed February 8, 2018
- Gill J, Dong J, Rose C, Johnston O, Landsberg D, Gill J: The effect of race and income on living kidney donation in the United States. J Am Soc Nephrol 24: 1872–1879, 2013
- Gill J, Dong J, Gill J: Population income and longitudinal trends in living kidney donation in the United States. J Am Soc Nephrol 26: 201–207, 2015
- Saad L: U.S. workers still haven't shaken the job worries of 2009. Available at: http://news.gallup.com/poll/164222/workers-havenshaken-job-worries-2009.aspx. Accessed January 29, 2018
- Bloembergen WE, Port FK, Mauger EA, Briggs JP, Leichtman AB: Gender discrepancies in living related renal transplant donors and recipients. J Am Soc Nephrol 7: 1139–1144, 1996
- Tushla L, Rudow DL, Milton J, Rodrigue JR, Schold JD, Hays R; American Society of Transplantation: Living-donor kidney transplantation: Reducing financial barriers to live kidney donation-recommendations from a consensus conference. *Clin J Am Soc Nephrol* 10: 1696–1702, 2015
- Ruck JM, Holscher CM, Purnell TS, Massie AB, Henderson ML, Segev DL: Factors associated with perceived donation-related financial burden among living kidney donors. Am J Transplant 18: 715–719, 2018

- Rodrigue JR, Schold JD, Morrissey P, Whiting J, Vella J, Kayler LK, et al.: KDOC Study Group: Direct and indirect costs following living kidney donation: Findings from the KDOC study. Am J Transplant 16: 869–876, 2016
- Gill JS, Gill J, Barnieh L, Dong J, Rose C, Johnston O, et al.: Income of living kidney donors and the income difference between living kidney donors and their recipients in the United States. Am J Transplant 12: 3111–3118, 2012
- Women's Bureau: Working mothers issue brief U.S. department of labor, 2016. Available at: https://www.dol.gov/wb/resources/WB\_WorkingMothers\_508\_FinalJune13.pdf
- Gifford K, Kessler J, Merion R, Mathur A, Ojo A: Effect of Lost Wage Reimbursement to Kidney Donors on Living Donation Rates, Ann Arbor, MI, Arbor Research Collaborative for Health, 2017
- 22. Ommen ES, Gill JS: The system of health insurance for living donors is a disincentive for live donation. *Am J Transplant* 10: 747–750, 2010
- Rodrigue JR, Fleishman A: Health insurance trends in United States living kidney donors (2004 to 2015). Am J Transplant 16: 3504–3511, 2016
- Percheski C: Men as Dependents? Marriage and Changes in Health Insurance Coverage among Working-age Adults in the United States, 1988 to 2008. Socius 3: 1–13, 2017
- Living donor protection act, United States 115th congress (2017). Available at: https://www.congress.gov/bill/115th-congress/housebill/1270/text. Accessed February 3, 2018
- Delmonico FL, Martin D, Domínguez-Gil B, Muller E, Jha V, Levin A, et al.: Living and deceased organ donation should be financially neutral acts. Am J Transplant 15: 1187–1191, 2015
- Hays R, Rodrigue JR, Cohen D, Danovitch G, Matas A, Schold J, et al.: Financial neutrality for living organ donors: Reasoning, rationale, definitions, and implementation strategies. *Am J Transplant* 16: 1973–1981, 2016
- Gill JS, Delmonico F, Klarenbach S, Capron AM: Providing coverage for the unique lifelong health care needs of living kidney donors within the framework of financial neutrality. Am J Transplant 17: 1176–1181, 2017
- Amaral S, Patzer RE, Kutner N, McClellan W: Racial disparities in access to pediatric kidney transplantation since share 35. J Am Soc Nephrol 23: 1069–1077, 2012
- 30. Mjoen G, Hallan S, Hartmann A, Foss A, Midtvedt K, Oyen O, et al.: Long-term risks for kidney donors. *Kidney Int* 27: 460, 2013
- Muzaale AD, Massie AB, Wang MC, Montgomery RA, McBride MA, Wainright JL, et al.: Risk of end-stage renal disease following live kidney donation. JAMA 311: 579–586, 2014

- Newell KA, Formica RN, Gill JS, Schold JD, Allan JS, Covington SH, et al.: Integrating APOL1 gene variants into renal transplantation: Considerations arising from the American society of transplantation expert conference. *Am J Transplant* 17: 901–911, 2017
- Kopp JB, Smith MW, Nelson GW, Johnson RC, Freedman BI, Bowden DW, et al.: MYH9 is a major-effect risk gene for focal segmental glomerulosclerosis. Nat Genet 40: 1175–1184, 2008
- 34. Tzur S, Rosset S, Shemer R, Yudkovsky G, Selig S, Tarekegn A, et al.: Missense mutations in the APOL1 gene are highly associated with end stage kidney disease risk previously attributed to the MYH9 gene. *Hum Genet* 128: 345–350, 2010
- Genovese G, Friedman DJ, Ross MD, Lecordier L, Uzureau P, Freedman BI, et al.: Association of trypanolytic ApoL1 variants with kidney disease in African Americans. *Science* 329: 841–845, 2010
- Reeves-Daniel AM, DePalma JA, Bleyer AJ, Rocco MV, Murea M, Adams PL, et al.: The APOL1 gene and allograft survival after kidney transplantation. Am J Transplant 11: 1025–1030, 2011
- Ibrahim HN, Foley RN, Reule SA, Spong R, Kukla A, Issa N, et al.: Renal function profile in white kidney donors: The first 4 decades. J Am Soc Nephrol 27: 2885–2893, 2016
- Ibrahim HN, Berglund DM, Jackson S, Vock DM, Foley RN, Matas AJ: Renal consequences of diabetes after kidney donation. Am J Transplant 17: 3141–3148, 2017
- Kumar K, Holscher CM, Luo X, Garonzik Wang J, Anjum S, King EA, et al.: Persistent regional and racial disparities in nondirected living kidney donation. *Clin Transplant* 31: 2017
- Axelrod DA, Guidinger MK, Finlayson S, Schaubel DE, Goodman DC, Chobanian M, et al.: Rates of solid-organ wait-listing, transplantation, and survival among residents of rural and urban areas. JAMA 299: 202–207, 2008
- Tonelli M, Klarenbach S, Rose C, Wiebe N, Gill J: Access to kidney transplantation among remote- and rural-dwelling patients with kidney failure in the United States. JAMA 301: 1681–1690, 2009

This article contains supplemental material online at http://jasn.asnjournals. org/lookup/suppl/doi:10.1681/ASN.2017111160/-/DCSupplemental.

See related editorial, "Gender Disparities and Financial Barriers to Living Kidney Donation," on pages 1081–1083.